Author's response to reviews

Title: Tailored implementation for chronic diseases (TICD): A protocol for process evaluation in five cluster randomized controlled trials in five European countries

Authors:

Cornelia Jaeger CJ (cornelia.jaeger@med.uni-heidelberg.de)
Tobias Freund TF (tobias.freund@med.uni-heidelberg.de)
Jost Steinhæuser JS (jost.steinhæuser@med.uni-heidelberg.de)
Eivind Aakhus EA (eivind.aakhus@gmail.com)
Signe Flottorp SF (Signe.Flottorp@kunnskapssenteret.no)
Maciek Godycki-Cwirko MGC (maciekgc@uni.lodz.pl)
Jan van Lieshout JVL (j.vanLieshout@iq.umcn.nl)
Jane Krause JK (jk208@leicester.ac.uk)
Joachim Szecsenyi JS (joachim.szecsenyi@med.uni-heidelberg.de)
Michel Wensing MW (M.Wensing@iq.umcn.nl)

Version: 2 Date: 19 August 2013

Author's response to reviews:

Dear Sir or Madam,

we are happy to submit our revised manuscript, which is now following the structure of a methodology paper.

Please note that this manuscript is part of the proposed series of manuscripts from the European project “Tailored implementation in chronic illness care” (contact person: Michel Wensing).

If you have further questions please do not hesitate to contact me or Mr. Wensing.

Yours sincerely,

Cornelia Jaeger